Author's response to reviews

Title: Eosinophilic myocarditis mimicking acute coronary syndrome secondary to idiopathic hypereosinophilic syndrome: a case report

Authors:

Reza Amini (aminim@ccf.org)
Craig Nielsen (aminim@ccf.org)

Version: 3 Date: 29 January 2010

Author's response to reviews:

Dear editorial team,

Thanks for reviewing our paper. Please find attached the finalized copy of our manuscript with following changed:

1- Reference number 20 citation was missing in the text. Citation is inserted for the table 1.
2- Comments made by the copyeditors are addressed: (Co-author's email address is corrected, and patient's nationality is added to the manuscript).
3- Highlighting is removed from the manuscript.
4- Visible vertical lines are removed from the table.

Thanks for publishing our manuscript

Regards,

Reza Amini, MD